Author's response to reviews

Title: Association between tobacco use and body mass index in urban Indian population: implications for public health in India.

Authors:

Mangesh S Pednekar (pednekarmangesh@rediffmail.com)
Prakash C Gupta (pcgupta@healis.org)
Heema C Shukla (Heema.shukla@hillingdon.nhs.uk)
James R Hebert (jhebert@sc.edu)

Version: 5 Date: 28 January 2006

Author's response to reviews: see over
Dear Dr Tony,

We have checked and revised our manuscript as per your suggestions.

We ensure that Title and Abstract are identical in the manuscript and in the submission system.

The list of authors is written identically and in the order in which they are listed in the manuscript system.

We have removed underline from the section headings.

Table 1, 2, 3, 4, and 5 oriented in portrait.

Thanks once again.

Mangesh S Pednekar